Author's response to reviews

Title: The context and practice of handwashing among new mothers in Serang, Indonesia: a formative study

Authors:

Katie Greenland (katie.greenland@lshtm.ac.uk)
Endang Iradati (eiradati@jhpiego.net)
Abigail Ati (abby.ati@gmail.com)
Yanti Yulianti Maskoen (butimaskoen@yahoo.com)
Robert Aunger (Robert.aunger@lshtm.ac.uk)

Version: 2 Date: 14 December 2012

Author's response to reviews: see over
13th December 2012

Dear Editor,

We are pleased to have been invited to resubmit our manuscript entitled “The context and practice of hand-washing among new mothers in Serang, Indonesia” to BMC Public Health. We are grateful to the two reviewers for providing in-depth and useful comments. We have taken these comments on board and incorporated suggestions as appropriate. See our detailed response to each comment for more information on how individual comments have been addressed.

We look forward to a favourable review of our revised manuscript.

Yours sincerely,

Katie Greenland
Research Fellow
Department of Disease Control